Author’s response to reviews

Title: Latent class analysis of sexual health markers among men and women participating in a British probability sample survey

Authors:

Alison Parkes (alison.parkes@glasgow.ac.uk)
Michael Waltenberger (michael.waltenberger@hotmail.com)
Catherine Mercer (c.mercer@ucl.ac.uk)
Anne Johnson (anne.johnson@ucl.ac.uk)
Kaye Wellings (kaye.wellings@lshtm.ac.uk)
Kirstin Mitchell (kirstin.mitchell@glasgow.ac.uk)

Version: 2 Date: 13 Nov 2019

Author’s response to reviews:

Response to Editor Comments:

1. Abbreviations
   Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

   Response: a list of Abbreviations has been provided.

2. Ethics approval and consent to participate
   In the Ethics approval and consent to participate statement of the Declarations, please state whether ethics approval was obtained.

   In this section, please also confirm whether informed consent was obtained from all participants and specify whether the consent was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

   Response: this information has been added.

3. Funding
   Please combine the Funding and Role of the funding source statements into the same section, titled, ‘Funding’.
Response: the two sections have been combined.

4. Authors’ contributions – missing authors CM, AJ, KW.
We note that the contributions of author CM, AJ and KW are missing from the Authors’ contributions statement of the Declarations. The individual contributions of ALL authors to the manuscript should be specified in this section. Guidance and criteria for authorship can be found in the online journal submission guidelines.

Response: the contributions of these authors have been clarified by using specific initials, as requested.

5. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: clean files have been uploaded.